MOSS Minutes

March 2, 2004

ELEMENT ATTENDANCE:

ESDIS SSDO and System Manager, MRTT, EDOS, EMOS, ASDC DAAC, OMI SIPS, Averstar, GES DAAC, HIRDLS SIPS, MLS SIPS, TES SIPS

NEXT TELECONS: March 16, 2004 at 2pm EST in Bldg 32 room S241

DIAL-IN NUMBER:

USA Toll Free Number: 866-581-4760

USA Toll Number: +1-210-280-9695 (international)

PASSCODE: 574686

MOSS 4 (5 day test) Data Preparation Schedule (Based on a LRD of no earlier than June 17, 2004):

· March 12: Write/verify tapes for EDOS local playback complete 

· New: March 16: One-day MOSS 4 pre-test

· March 22-26: MOSS 4 dry run 

· March 30: MOSS 4 TRR 

· April 5-10: Formal MOSS 4 run for the record 

· Update: April 14: Requirements status update complete 

· Update: April 19: MOSS 4 Debrief 

· Update: April 20: Aura SORR Dry Run (via telecon)
· New: April 26: Aura MRR (invitation only)
· May 3: Final slides for formal SORR presentation complete 
· May 11: Aura SORR formal presentation (at GSFC)
MOSS 4 Telecon Schedule (all telecons Tuesday at 2:00 PM EST): 

March 16, March 30 (TRR), April 13 (cancelled), April 20 (Debrief)
MEETING HIGHLIGHTS:
Agenda

1. Karen mentioned that at the Aura FOR last week, ESDIS received a RFA to reconsider holding the Aura SORR before the MRR (April 26th). Karen mentioned that in order to do that she would have to move up MOSS 4, which is too risky to do at this point. Karen recommended to Glenn (for discussion with Senior Staff) that we keep MOSS 4 as is but have the MOSS 4 Debrief on April 19th and the SORR Dry Run April 20th.  After the telecon, Karen sent an email to the Aura Test list on March 4th with the proposed schedule changes, requesting any issues with the changes to be sent to her by March 8th.

2. New Schedule Item – Proposing Pre Dry Run—one-day test-- for week of March 15th (preferable Tuesday the 16th).  Purpose to check out new data and secure interfaces

Karen stated that the purpose of this test is to perform a quick checkout of the data and the secure interfaces; to ensure that everything is okay. A lot of changes were made to the data. It was decided to hold the test on Tuesday March 16th. The test would also validate MLS and TES computer room move, ASDC ( JPL performance, and any SIPS upgrades. This test would be in additions to the Dry Run starting March 22nd.  All teams expressed their ability to support the one-day test.  Karen mentioned that any verification done with the data would be greatly appreciated.

Susan of MRTT mentioned that Mission Rehearsal is also scheduled for that week and does not know if this test would impact EMOS. EMOS would need to distribute carryout files to the GES DAAC. Karen will talk to Dan/Hillary, if a problem, the carryout files delivered to the GES DAAC last Friday (2/27/04) for GBAD test, could be used.

Nancy is to double check with Ron regarding EDOS’ ability to support the test on March 16th.  After the telecon, Nancy checked with Ron and EDOS will be able to support the test.

3. Status of MOSS 4 Data

· GBAD

· EDSs for secure interface testing

· MLS; HIRDLS; TES

Bill mentioned that GBAD for MOSS 4 is done; processed by GES DAAC and distributed to the SIPS and ASDC DAAC yesterday (3/1/04). The GBAD for the TES special products has been given to Eric for tape creation but the data appears to have the incorrect data date. Eric is awaiting a new delivery before the tape is made. Once the tape is created and given to EDOS, the data would be sent to EMOS for carryout files generation, then sent to the GES DAAC to produce attitude output from DPREP, and finally to ASDC DAAC for distribution to TES SIPS. Bill mentioned that this could happen this week. But TES mentioned that they would be down tomorrow for the move. Karen asked them to let her know when they are back up and ready to receive this DPREP data. TES already has the definitive ephemeris output from DPREP. The TES point of contact for this data is Scott Lewicki.

Bill mentioned that ESDIS received the new data (MOSS 4) from HIRDLS and confirmed with TES that the data that ESDIS currently has will be used.

4. Status of ASDC DAAC --> JPL (TES/MISR) network traffic

· Status of NCR fix for MISR

· Performance retest

Steve reported that there has been no change since Friday (ASDC DAAC telecon). Steve had conversations with Bryon that Friday afternoon and Bryon feels that he knows what he left out and would try to have it to ASDC by Thursday (3/4/04) in time for their system reboot.

5. Status of Secure Ingest and Distribution testing

· ASDC DAAC/TES SIPS

· Status of TT submitted for slow secure transfers (ftp vs. scp)

· Status of remaining work

· GES DAAC/ HIRDLS, MLS, OMI SIPS, ODPS

· Status of remaining work

· Status of NCR for special characters in password

· Key Exchange

· Set-up @ DAAC?

· Set-up @ SIPS  - all or ODPS only?

ASDC DAAC Secure Ingest and Distribution status – Steve reported that secure ingest and distribution functions are in OPS and the concern he has is with the throughput. As soon as confirmation is received from TES SIPS, Steve will change from ftp to scp. MISR production is not at 5X yet, so cannot track a lot of performance yet. Steve mentioned that the test on March 16th would be a good time to do the performance test.

GES DAAC Secure Distribution / Remaining Work status – GES DAAC is waiting to hear from Mark Echeverri in order to start secure distribution testing in TS2 mode followed by OPS. Audrey mentioned that MLS SIPS has decided to go with F-secure. David mentioned that the main reason was that the data would not be encrypted.

Ed also mentioned that he did not notice any degradation in performance during the CPT6 data delivery test via secure transfers. For example, a 324 MB file took 4 minutes. They are waiting to hear from Bryon for the re-scheduling of the key exchange work. He also mentioned that he had not seen the fix for NCR 39016 (special characters in the password), but it is being looked into. After the telecon, Latha informed Karen that Bryon came over to the DAAC on 3/3/04 to help with the key exchange. Progress was made.

6. Status of HIRDLS time jump fix

Ken mentioned that they are in the architecture design phase.

7. Status of Engineering Testing with GES DAAC

· MLS (L2s, L3s, QA, PH, DAP)

· HIRDLS (HIR1CAL – EDG)

Audrey mentioned that a lot of this testing depends on the completion of F-secure implementation. She would like to do some interface testing with the DAAC prior to their move (computer room). She also mentioned that they have the new DPREP products (GBAD test).

Ed of the GES DAAC mentioned that the HIRCAL1 ESDT has to be promoted to OPS. The latest version is in TS2 and has been since mid December 2003. During the telecon, Ed verified that the HIRCAL1 product received from HIRDLS in February 2004 did indeed have the start and stop time fields populated. This would be re-verified in OPS once the updated ESDT is promoted.

8. Status of OMI SIPS-ODPS MT#2 re-run

· Status of L2 ESDTs (OMI SIPS and GES DAAC)

· ESDTs promoted to OPS?

Elaine reported that the OMI SIPS-ODPS MT#2 re-run is complete. They generated all of the L1B products and are trying to produce L2s. The testing is currently in GES DAAC TS2 mode.

9. Status of TES SIPS Release 5

· Engineering test with ASDC

TES SIPS mentioned that they have all the Release 5 PGEs and things are going okay. They are not planning any engineering test with ASDC prior to MOSS 4 other than the MOSS 4 dry runs.

10. Status of EDOS-EMOS Secure Transfer


- Test schedule?

Nancy of EDOS reported that several tests were conducted in which the interfaces were exercised: 2/25/04; GBAD test on 2/27/04 and the CPT6 data delivery on 03/01/04.

11. SCIF 4 Detail Science Debrief to be held late March – need inputs by mid March; detail questions will be sent out very soon

Karen mentioned that she believe the Detail Science Debrief is going to be the week of March 25th. She will send out an email with the detail questions. After the telecon, Karen sent an email on March 4th with the information required from the teams in preparation for the SCIF 4 Science Debrief.

12. CPT 6 Data Delivery – March 1st
· Any issues?

No CPT 6 data delivery issues reported.

13. Aura FOR RFA’s and Aura MRR scheduled for April 26th
Schedule for MOSS 4 requirements status and debrief may need revised
See # 1 above.

ACTION ITEMS: Documented in the Aura Action Item Tool at http://jupiter02.gsfc.nasa.gov/auratesting/.  Please remember to enter your status in the tool.
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